Dear Editor,

We are resubmitting our manuscript "A case-control study of physical activity patterns and risk of non-fatal myocardial infarction". As requested we have made the following changes:

1) We have added the competing interests section
2) The ethics statement can be found at the end of the Study population subsection in the Methods section and it says "Cases and controls provided informed consent on documents approved by the Human Subjects Committee of the Harvard School of Public Health and the University of Costa Rica." Please, let us know if this is a valid ethics statement, and if not, what else we have to add.
3) We have added the aims of the study in the Background section of the Abstract

Thank you for your consideration.

Your sincerely,

Ana Baylin